Author's response to reviews

Title: Engineered Measles Virus Edmonston Strain Used as a Novel Oncolytic Viral System Against Human Hepatoblastoma

Authors:

Shucheng ZHANG (zhangshucheng76@126.com)
Weilin WANG (zhangshucheng2009@163.com)
Weisong CAI (cailab@126.com)
Kailei JIANG (nancy@sina.com)
Zhengwei YUAN (yuanzw@163.com)

Version: 3 Date: 16 April 2012

Author's response to reviews: see over
Shucheng Zhang M.D.
Department of Pediatric Surgery, Major Laboratory of Chinese Health Ministry for Congenital Malformations
The Shengjing Hospital, China Medical University
36 SanHao Street, Heping District, Shenyang, Liaoning, 110004 P.R.China
April 16, 2012

BMC Cancer

Dear editor:

I submitted the manuscript titled “Engineered Measles Virus Edmonston Strain Used as a Novel Oncolytic Viral System Against Human Heptoblastoma.” to be considered for publication as an “Research” in the journal of BMC Cancer. All data contained are accurate and all statements asserted as facts are based on careful research by the authors. There has been no duplicate publication or submission elsewhere. There is no ethical problem or conflict of interest.

We certify that we have participated sufficiently in this work to take public responsibility for the content of this manuscript; believe the experimental design and method as well as the collection, analysis, and interpretation of the data are sound; and have reviewed the final version of the manuscript and approve it for publication. The correspondence author was for material collection and the first author was for manuscript written and translation;

Correspondence should be directed to Wei-Lin WANG at the following address:
Wei-Lin WANG MD.PhD
Department of Pediatric Surgery
Major Laboratory of Chinese Health Ministry for Congenital Malformations
Shengjing Hospital of China Medical University
36 SanHao Street, Heping District, Shenyang, 110004, China.
Ph: 86-24-28851271
Fax: 86-24-24834499
E-mail: zhangshucheng2009@163.com

Thanks very much for your attention to our paper.

Very sincerely yours,
Shucheng ZHANG M.D.